CORBA OODS I Tutorial 1

© 2001, Uwe Rastofer Universität Erlangen Nürnberg, Informatik 4

Reproduktion oder Verwendung dieser Unterlage zu Lehrzwecken außerhalb der Universität Erlangen ist ohne Genehmigung der Autoren unzulässig

Homework Assignment #10: Java Client (Due Jan. 18, 2001 9:30) 0 points

Nowyouhave to implementa clientGUIforthe librarysystem. Youwilluse ORBacus for Java 4.0.3forthe client.AllORBacusbinariesare in/local/ORBacus-4.0.3/bin and you have to add /local/ORBacus-4.0.3/lib/OB.jar to yourClassPath. On LInux you also have to set the environment variable LD_LIBRARY_PATH of your shell to /

local/ORBacus-4.0.3/lib to make the IDLcompilerwork.Ensure that youuse JDK

1.3.x(e.g. from/local/java-1.3) because ORBacus 4.0 does not workcorrectlywith earlier JDK versions. It is also a good idea to make a copy of the Java client in /proj/i4oods/

hello_java_client and test it before you start.

a) Use the IDLdefinitionsfrom/proj/i4oods/pub/assignment10/Library.idl.

These are the same interfaces as inthe second solutionofAssignment#9. Tocompile the IDLfileuse jidl withthe option --package generated. This willgenerate portable stubs (and server side skeletons, which you will not need at the moment).

b) A server implementationof the librarysystem is provided for you. You willfind a reference to a Librarian object inthe file/proj/i4oods/pub/assignment10/

Librarian.ior. Your client should read this file at every startup and obtain the

reference fromit.(IfIrestart the test server, thisfilechanges, so don’tcopyitintoyour directory.)

c) Yourclienthas to providea TextField to enter the name ofa card owner. Withtwo buttons you can either find an existing card or issue a new card.

d) In another TextField the user may enter a search term. The client should then obtain a list of matching books.

e) To displaythe listofmatchingbooks you can either show informationon one book at a time(e.g. in a TextArea) and providebuttons to get to the next or previous book. Another way is to display aList of all books.

f) Provide a button to borrow the currently displayed book from the list in item (e). g) Show the same kindoflistas initem(e) forthe books that are currentlyborrowedon the card owner obtained in item (c).

h) Provide a button to return the currently displayed book from the list in item (g). i) Keep in mind that remote methods may also throw CORBA System Exceptions.

Youmay use a Java GUIBuilderlikeJava Workshopor NetBeans instead ofcodingthe GUI by hand. It also does not matter if you use AWT or Swing for the GUI.